

JGR Solid Earth

RESEARCH ARTICLE

10.1029/2020JB019745

Key Points:

- New paleomagnetic data from Pico Island in the mid-North Atlantic region
- Data reveals an inclination low with minimum values of 32° from 900 to 1560 AD
- We present the first full-vector paleosecular variation curve for the Azores covering the past 2 kyr, suitable for paleomagnetic dating

Supporting Information:

- Supporting Information S1
- Data Set S1
- Table S1

Correspondence to: A. Béguin, annemarieke.beguin@ntnu.no

Citation:

Béguin, A., Pimentel, A., & de Groot, L. V. (2021). Full-vector paleosecular variation curve for the Azores: Enabling reliable paleomagnetic dating for the past 2 kyr. *Journal of Geophysical Research: Solid Earth*, *126*, e2020JB019745. https://doi. org/10.1029/2020JB019745

Received 9 MAR 2020 Accepted 20 DEC 2020

© 2021. The Authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

Full-Vector Paleosecular Variation Curve for the Azores: Enabling Reliable Paleomagnetic Dating for the Past 2 kyr

Annemarieke Béguin¹, Adriano Pimentel^{2,3,4}, and Lennart V. de Groot¹

¹Paleomagnetic Laboratory Fort Hoofddijk, Utrecht University, Utrecht, Netherlands, ²Centre for Information and Seismovolcanic Surveillance of the Azores (CIVISA), Ponta Delgada, Portugal, ³Research Institute for Volcanology and Risk Assessment (IVAR), University of the Azores, Ponta Delgada, Portugal, ⁴Research Centre in Biodiversity and Genetic Resources (CIBIO), InBIO Associate Laboratory, Pole of the Azores, Ponta Delgada, Portugal

Abstract For archeomagnetic dating, high-quality directional and intensity paleosecular variation curves are needed. The Azores Archipelago in the mid-Atlantic Ocean provides a wealth of volcanic products erupted during the Holocene, making it an ideal location to (1) gather paleomagnetic data from well dated lava flows and (2) construct a paleosecular variation (PSV) curve that enables paleomagnetic dating of volcanic products with unknown age. Here, we present new full-vector paleomagnetic data from Pico Island, and combine the new data with existing data from neighboring islands to construct a new full-vector PSV curve for the Azores Archipelago. An extensive rock-magnetic study underpins the quality of our paleomagnetic carriers. From Pico Island, we obtained 21 new mean site directions; and 15 paleointensity estimates with the multimethod paleointensity approach from 12 sites, the age was known for 14 and 10 sites, respectively. By bootstrapping the non-Gaussian uncertainty estimates of the radiocarbon age calibrations and the confidence intervals associated with the direction and paleointensity estimates, we produce the first full-vector PSV curve with confidence intervals for the Azores covering the past 2 kyr. The PSV curve reveals a period of low inclination between ~900 and 1560 AD, with minimum values of 32°. The potential of our new full-vector PSV curve is demonstrated by successfully dating five lava flows from Pico Island.

Plain Language Summary The Earth's magnetic field is highly variable over time and space. Lava flows archive the geomagnetic field upon emplacement and thereby forming an archive of geomagnetic field changes. The Azores Archipelago is located in the mid-North Atlantic Ocean providing a unique location to study the changes of the Earth's magnetic field. Here, we present new paleomagnetic data from Pico Island, Azores. We combine the newly derived data with available data from neighboring islands and compose a curve of the changes, for both the direction and the strength of the Earth's magnetic field over the past 2,000 years.

1. Introduction

The Azores Archipelago is a group of nine oceanic volcanic islands located in the North Atlantic Ocean. Due to its particular geodynamic setting resulting from a triple plate junction, the archipelago is subjected to frequent volcanic and seismic activity (Gaspar et al., 2015). Over time it produced a wealth of volcanic rocks, archiving geomagnetic field variations. This gives the unique opportunity to study the changes in the geomagnetic field throughout time in the middle of the North Atlantic Ocean, a remote area that would otherwise be poorly constrained in geomagnetic field models as the Earth's magnetic field is highly variable in time and space. Regional geomagnetic curves can be used for the dating of volcanic products (e.g., Greve et al., 2016) and are preferred over global models as regional changes of the field might be smoothed in global geomagnetic field models (Pavón-Carrasco et al., 2011). For areas with high data density, paleosecular variation (PSV) curves can be used to describe the field behavior for small areas or at remote locations. And high-quality PSV curves can also be used as a tool to date recent volcanic units, ultimately contributing to improve volcanic hazard assessment in active volcanic regions (e.g., Böhnel et al., 2016), such as the Azores (Di Chiara et al., 2014a) and are vital to further our understanding of the short term behavior of the geodynamo.

To construct a PSV curve, high-quality paleomagnetic data are required. Obtaining a reliable direction (declination and inclination) of the paleofield is relatively easy. Reliable estimates of the past strength (the paleointensity) of the Earth's magnetic field, however, are often hampered by thermal or chemical alteration of the magnetic signal during the experiments. Previous studies in the Azores Archipelago derived high-quality full-vector data for the islands of São Miguel (Di Chiara et al., 2012, 2014b) and Terceira (de Groot et al., 2016). Di Chiara et al. (2012) presented the first directional PSV curve spanning the last 3 kyr for the Azores based on 13 well dated flow-mean directions from São Miguel Island, including directional results from a previous study by Johnson et al. (1998). This curve, however, is poorly constrained prior to 750 AD due to the low data density. More recently, de Groot et al. (2016) derived high-quality full-vector data from Terceira Island further contributing to the paleomagnetic data set of the Azores. They applied a multimethod paleointensity approach and obtained results for 60% of all sampled cooling units.

Here, we present new full-vector paleomagnetic data from Pico Island, from 22 sites sampled from 21 different cooling units. Samples from all sites were subjected to an extensive rock-magnetic study, including susceptibility and magnetization as function of temperature, hysteresis analyses, and scanning electron microscopy to test their suitability as recorder of the geomagnetic field. Thermal and alternating field demagnetization experiments were performed to obtain paleodirections. New paleointensity estimates from Pico Island were derived with the multimethod paleointensity approach, following de Groot et al. (2013).

Together with the available data from São Miguel and Terceira Islands, we present the first full-vector PSV curve for the Azores region for the past 2 kyr. The input data set for our new Azores-PSV-curve only considers independently dated sites (albeit historical sources or radiocarbon analyses), and consists of: 12 mean directions, and 8 paleointensity estimates from São Miguel, 4 mean directions, and 6 paleointensity estimates from Terceira, and the newly derived 14 mean directions and 13 paleointensity estimates from Pico Island for the past 2 kyr. We bootstrapped the data to take the uncertainty of the paleomagnetic data (direction and intensity) and the non-Gaussian age error distribution of the calibrated radiocarbon ages into account. This resulted in a full-vector PSV curve with confidence intervals that is successfully used for constraining the ages of five recent lava flows from Pico Island.

2. Geological Setting and Sampling

The nine volcanic islands of the Azores Archipelago (36.9°N-39.8°N, 24.9°W-31.3°W) straddle the triple junction of the North-American, Eurasian, and Nubian lithospheric plates in the mid-North Atlantic Ocean (inset of Figure 1). Pico Island (447 km²) has an elongated shape with a maximum length and width of 46 and 16 km, respectively. It is the youngest island of the Azores Archipelago (270 ± 150 ka; Demande et al., 1982) and is composed of three volcanic systems: Topo volcano, Planalto da Achada fissure zone and Pico volcano (Figure 1). Topo is an extinct volcano located in the southern part of the island that is deeply eroded and covered by younger lavas. Planalto da Achada fissure zone corresponds to a WNW-ESE-oriented ridge of scoria cones, ~30-km long, that forms the eastern part of the island. Pico volcano is the youngest volcano of the archipelago (53 ± 5 ka; Costa et al., 2014) and constitutes the western part of the island. This central volcano is characterized by gentle lower slopes with numerous scoria cones, rising up to 2,351 m as a conical shape edifice with steep slopes toward the top (Nunes, 1999, 2020; Zanon et al., 2020). The geological record of Pico Island reveals a high eruptive frequently during the Holocene (Nunes, 1999). Planalto da Achada fissure zone erupted at least 13 times in the last 2,000 years and Pico volcano erupted 22 times in the last 1,500 years (Nunes, 1999). After the island became inhabited in the 15th century, three historical eruptions have been recorded (Nunes, 1999; Zanon et al., 2020): Planalto da Achada fissure zone in 1562-1564 AD (Mistério da Praínha); Pico volcano in 1718 AD along two subaerial eruptive centers (Mistério de Santa Luzia on the north flank and Mistério de São João on the south flank) and a submarine vent offshore the south coast, and again in 1720 AD (Mistério da Silveira).

We sampled 22 sites from 21 independent cooling units on Pico Island (Figure 1 and Table 1). For each site, \sim 10 oriented samples were taken up to 20 m apart across each cooling unit to obtain a reliable paleodirection. For site PI07, the samples were taken closer together as the outcrop was rather small. Standard size paleomagnetic cores (2.5 cm in diameter and up to 15 cm in length) were collected using a petrol-powered





Figure 1. Location of sampled sites (red icons) on simplified geological map (modified after Nunes (1999)), lava flows are colored by the respective age rages; same colors are within same age boundaries. The three volcanic systems are indicated by the black solid lines. Inset shows the geographic position of Pico Island (red rectangle) in the Azores Archipelago and in the North Atlantic Ocean, the main structural features (NAP, North-American plate; MAR, mid-Atlantic Ridge; TR, Terceira Rift; EAFZ, East Azores Fracture Zone; GF, GLORIA Fault), and the locations of Terceira and São Miguel Islands (blue dashed circle and oval).

drill. Age constraints for most of these sites were provided by either historical records (PI01, PI02, PI03, PI04, PI07) or radiocarbon dating of charcoal, turf material, and organic sediment by Nunes (1999). We recalibrated the radiocarbon ages using the IntCal13 calibration curve (Reimer et al., 2013) and the MatCal software package (Lougheed & Obrochta, 2016) (Table 1). Six sites (PI10, PI17, PI18, PI20, PI21, PI22) were sampled from lava flows with unknown ages; however, a relative age estimate can be constrained from the geological map (Figure 1). These six sites are to be dated with our new PSV curve.

3. Rock-Magnetic Behavior and Microscopic Characterization

To assess the suitability of the lava flows for paleodirection and paleointensity experiments, we performed both high-field and low-field rock-magnetic analyses on one sample per site. The susceptibility and magnetization as function of temperature were measured to check for chemical alteration and to determine the Curie temperatures (T_c). Together with the thermal demagnetization behavior, this was used to optimize the temperature steps in the thermal paleointensity experiments (Figure 2). For each site, the susceptibility was measured as function of temperature on an AGICO KLY-3 susceptometer with a CS-3 furnace attachment. The samples were heated in seven cycles with approximate peak temperatures of 190, 250, 310, 370, 430, and 590 °C; after each peak the temperature was lowered 50 °C to test for the reversibility of the signal, i.e., test for chemical alteration of the sample. The highest temperature that shows reversible behavior is referred to as the alteration temperature (T_{alt}), this is the highest temperature that can safely be used in paleointensity



Ta	ble	1	

Sites and Sampling

	1 0						
Site (PI)	Location	n [UTM]	Age	Age constraint	Calibrated radiocarbon age (AD)	One sigma interval [lower bound-upper bound, relative probability]	Dating site
1	389802	4260193	_	_			
2	388632	4257891	1563 AD	Historical			
3	385757	4253650	1720 AD	Historical			
4	380375	4252736	1718 AD	Historical			
5	382440	4254404	$625 \pm 65 \text{ BP}$	¹⁴ C	1346	[1292–1328, 0.392; 1341–1395, 0.608]	CP15
6	382264	4266554	$1725 \pm 55 \text{ BP}$	¹⁴ C	310	[252-305, 0.437; 311-383, 0.563]	CP06
7	374610	4268597	1718 AD	Historical			
8	370751	4266356	$1615 \pm 140 \text{ BP}$	¹⁴ C	423	[258-284, 0.071; 322-590, 0.929]	CP05
9	372469	4261311	$1390 \pm 70 \text{ BP}$	¹⁴ C	641	[575-687; 1.00]	CP13
10	369502	4261662	—	—			
11	379341	4256320	$1310 \pm 70 \text{ BP}$	¹⁴ C	719	[652–771; 1.00]	CP10
12	379085	4256245	$1670 \pm 115 \text{ BP}$	¹⁴ C	368	[244-436, 0.756; 446-472, 0.08; 486-535, 0.164]	CP11
13	382093	4260656	$365 \pm 75 \text{ BP}$	¹⁴ C	1543	[1453-1524; 0.502]	CP09
14	395247	4255968	$2720 \pm 50 \text{ BP}$	¹⁴ C	-873	[905 BC-820 BC; 1.00]	CP19
15	398717	4255119	$1405 \pm 50 \text{ BP}$	¹⁴ C	630	[602-663; 1.00]	CP21
16	397754	4256649	$1105 \pm 45 \text{ BP}$	¹⁴ C	934	[893–986; 1.00]	CP20
17	409939	4252073	_	_			
18	406651	4255178	—	—			
19	395317	4250315	$3520 \pm 60 \text{ BP}$	¹⁴ C	-1844	[1921 BC-1762 BC; 1.00]	CP17
20	402047	4251899	_	_			
21	373649	4253690	_	_			
22	380955	4264179	_	_			

For all sites the follow information is provided: site number; location in UTM coordinates (UTM zone 26S); its age as provided (uncalibrated laboratory age for the 14C dating); the nature of the age constraint (14C: radiocarbon; historical: constrained based on historical observations); the median probability of the radiocarbon age (if applicable) after calibration with the INTCAL.13 curve; the one sigma probability intervals (68.3%) obtained by the radiocarbon calibration; and the reference dating site from Nunes (1999). Sites PI01 and PI02 are from the 1562–1564 historical lava flow, we use a mean age of 1563 AD for the eruption.

experiments (Table 2). The magnetization over temperature was measured on a horizontal translation Curie balance (Mullender et al., 1993). The Curie temperature was determined with the two tangent method and alternation of the sample was again determined by the nonreversibility of the magnetic signal.

Samples representative of seven sites show at least two dominant Curie temperatures, for sites PI05, PI10, and PI17 the Curie temperatures cannot be determined as the magnetization over temperature results in a gradual decay. The alteration temperature obtained by the magnetization or susceptibility vs. temperature experiments differ slightly for sites PI02, PI04, PI05, PI07, PI08, PI09, and PI21 (Table 2).

Based on the rock-magnetic behavior of the susceptibility and magnetization as function of temperature, the sites are divided in rock-magnetic groups after de Groot et al. (2015, 2016) (Table 2). Rock-magnetic group L* is characterized by a sharp decrease in susceptibility from room temperature onward with a Hopkinson peak below room temperature. The remaining susceptibility at 150 °C is <80% of the room temperature susceptibility. Group H is the high-temperature group, it is characterized by an increasing or constant susceptibility from room temperature until at least 350 °C. Samples in group H retain 80% of their room temperature susceptibility at 400 °C, and have high Curie temperatures. Samples that show mixed behavior of both low and high Curie temperatures generally fall within group M, if the Hopkinson peak is below room temperature the samples are placed in group M*. Examples for each group are shown in Figure 2, results for all other sites are presented in supplementary information (S1). Only samples from sites PI11



Journal of Geophysical Research: Solid Earth



Figure 2. Rock-magnetic characterization. For each rock-magnetic group typical examples of susceptibility as function of temperature (a-d), magnetization as function of temperature (e-h), and Zijderveld diagrams (i-l) representing the thermal demagnetization behavior. The temperature $(^{\circ}C)$ steps and obtained directions are given in the Zijderveld diagrams.

and PI20 are categorized as rock-magnetic group L*. Group H contains seven samples from sites PI08, PI09, PI12, PI15, PI16, PI19, and PI21. Group M contains eight sites (PI01, PI02, PI04, PI06, PI07, PI17, PI18, and PI22) and M* five sites (PI03, PI05, PI10, PI13, and PI14).

The high-field magnetic behavior was measured on a MicroSense vibrating sample magnetometer (VSM), both hysteresis loops and back-field curves were measured with a maximum applied field of 1.5 T. The saturation magnetization (Ms), remanent saturation magnetization (Mr), coercive field (Bc), and remanent coercive field (Bcr) values were determined and the magnetic domain states were assessed from a Day plot (supplementary information S2). Samples from all sites plot within the pseudo-single-domain (PSD) range with (0.07–0.27) Mr/Ms ratios and (1.50–2.69) Bcr/Bc ratios and follow the mixing lines of Dunlop (2002), with the exception of site PI04 with a high Bcr/Bc ratio and plots outside the PSD range and closer to the multidomain (MD) range.

Samples from sites with known age were checked for oxidation and exsolution. Carbon coated thin sections were examined with a JEOL JCM-6000 table-top scanning electron microscope (SEM) in backscatter mode, using an acceleration voltage of 15 kV. The scanned samples were categorized using the oxidation classes as described by Watkins and Haggerty (1967) (Table 2; supplementary information S3). Twelve of the sixteen scanned samples (PI01–PI05, PI08, PI09, PI11–PI14, PI19) show little to no oxidation and exsolution and are placed in classes I, I–II, or II. These samples do not show abundant differentiated lamellae; however, the grains contain considerable amounts of holes and distortions. Only the sample from site PI12 is categorized as class I as it does not show holes but rather dendrite shaped grains. For sites PI06, PI07, PI15, and PI16 abundant lamellae are visible in the SEM images, placing them in oxidation class II–III and III.



MUCh	manne	Dools a	comotio	- nun	die Othe	tion t	Dol	00000	o ito an	dimontio	\$		E	The Towner				N. S.	T or or or o	"ollio"			È	dT ob too	11:04	
		RUCK-III	Iagueur	CIIAIA	CIEIIZA		Гà	PIIIOAI	SHELLC	nitectio			-	IEIIIAI-III	Iallia				Owave-L	Iallia			5	III-onna		
Site (PI)	Median age (AD)	T _c (°C)	$T_{\rm alt}$ (°C)	RT susc. $\times 10^{-6}$ $(m^{3/}$ kg)	CI.	Rock mag. group	dec (°)	(°)	α ₉₅ (°)	×	N (rej.)	N/u	n_{-}] int	Paleoint. (μT)	σ (μT) _I	$\sigma/$ paleoint	N/u	n P.	aleoint. (μT)	α μT) p	σ/ aleoint	N/u	n H int	aleoint. (μT)	σ (μT)]	$\sigma/$ paleoint
	I	180/560	400	3.25	I-II	M	16.2	56.1	5.1	103.7	6	4/5	65	41.8	24.0	57%	3/5	586	60.4	5.2	%6	4/6	65	54.5	13.7	25%
0	1563	250/560	550 (320)	2.46	II-II	М	-2.1	39.3	6.1	73.2	6	5/5	684	44.0	6.1	14%	6/6	410	44.1	9.2	21%	7/11	242	60.1	12.2	20%
3	1720	100	400	3.10	Π	W*	-9.5	64.0	5.8	79.1	10(1)	I		I	Ι	I	2/3	56	88.2	48.6	55%	4/8	48	61.9	17.6	28%
4	1718	100/510	375 (450)	10.40	Π	М	-17.1	62.7	2.8	328.0	6	2/5	34	64.7	27.3	42%	Ι	Ι	I	Ι	I	3/6	57	39.3	2.2	6%
2	1346	Gradual	350 (300)	3.50	II-II	W*	9.5	32.6	4.2	151.0	6	3/5	9	49.3	8.6	17%		I	I	I	I	3/10	31	59.7	6.0	10%
9	310	200/550	350	2.31	-II II	Μ	-0.8	51.5	4.6	126.2	6	7/7	235	71.3	26.0	37%	Ι	Ι	I	Ι	I	9/0	I	I	I	I
5	1718	250/550	200 (600)	13.80	Η	М	-8.0	66.0	4.8	115.0	10(1)	5/5	134	43.2	4.9	11%	1/1	7	185.9	Ι	I	4/6	109	55.1	15.5	28%
8	423	470	200 (475)	16.80	II-II	Н	I	I	I	I	I		I	Ι	I	I			I	I	I	9/0		I	I	I
6	641	450	450 (150)	15.20	Π	Н	19.6	45.0	4.1	187.4	×	5/5	171	36.8	18.1	49%	I	Ι	I	Ι	I	0/5	Ι	I	Ι	I
10		Gradual	350	5.43		W*	5.7	28.9	5.8	108.0	7				I	I			I			2/6	40	51.9	0.9	2%
11	719	80	400	1.50	II-II	Ľ*	18.1	56.8	4.1	182.9	×				I	I			I			4/6	161	69.4	3.2	5%
12	368	540	400	4.30	I	Н	-0.6	55.4	4.8	113.8	6	5/5	65	48.1	12.6	26%			I			2/6	24	60.0	10.4	17%
13	1543	150/580	350	2.07	II-II	W*	-0.9	34.1	3.9	179.1	10(1)				I	I	2/2	45	94.6	14.8	16%	6/10	158	41.9	5.4	13%
14	-873	530	320	1.92	II-II	¥	18.1	55.7	5.0	121.4	10(2)	6/9	70	62.4	17.7	28%			Ι			4/6	55	74.0	4.6	9%9
15	630	570	450	5.80	-H H	Н	15.1	55.6	2.9	271.5	10	5/5	282	67.8	10.9	16%	0/1	I	I	I	I	1/6	25	38.7	0.1	%0
16	934	530	420	9.25	III	Η	10.3	35.4	3.6	205.3	6	5/5	427	75.8	9.7	13%	5/6	525	59.4	8.5	14%	4/6	112	51.8	6.1	12%
17		Gradual	400	7.15		М	11.9	59.8	5.2	100.5	6				I	I	1/2	143	35.3	1.4	4%	9/0		I	I	I
18		220/550	350	1.88		М	2.7	57.0	4.4	119.5	10				I	I	4/6	95	64.9	47.4	74%	5/6	127	55.6	8.3	15%
19	-1844	580	475	7.05	II-II	Η	-15.8	55.7	3.6	203.5	6	3/8	29	31.2	1.4	4%	1/1	1	87.8			9/0		I		
20	Ι	100	350	4.17	Ι	Г*	-10.1	32	3.6	178.1	10	Ι	Ι	Ι	Ι	Ι	Ι	I	Ι	Ι	Ι	1/6	28	30.16	0.44	1%



4. Paleomagnetic Directions

Paleomagnetic directions were obtained from both thermal and alternating field demagnetization experiments. Four samples per site were thermally demagnetized in 13 temperature steps: 60, 100, 150, 200, 250, 300, 350, 400, 450, 500, 530, 560, and 600 °C, using an ASC TD48-SC furnace and were measured on a 2G cryogenic magnetometer (Figure 2, supporting information S4). For some samples, the magnetic moments at lower temperatures were beyond the dynamic range of the 2G cryogenic magnetometer; these low--temperature steps were not taken into account for the interpretation of the sample direction. Six to eight samples were demagnetized by increasingly higher alternating fields (AF) in 17 steps using a robotized 2G cryogenic magnetometer (Mullender et al., 2016); with peak fields of 2.5, 5, 7.7, 10, 15, 20, 25, 30, 40, 50, 70, 80, 100, 150, 225, and 270 mT. All demagnetization data were analyzed using paleomagnetism.org (Koymans et al., 2016), site mean directions are calculated using Fisher statistics (Fisher, 1953), and the Vandamme cutoff (Vandamme, 1994) was used to identify potential outliers (Table 2).

The obtained site means have small α_{95} values (<7°), and the lowest obtained *k* value is for site PI02 with a value of 73.2. The directions of sites PI01 and PI02 should be the same since both sites are from the 1562 to 1564 AD historical lava flow: PI01 was sampled on the northern branch, whereas PI02 was sampled on the southern branch. The directions of these two sites are, however, not the same and this deviation needs to be explained. The two sites sampled on the 1718 AD historical lava flows, PI04 on the southern flank of Pico volcano and PI07 on the northern flank, have a common true mean direction (Koymans et al., 2020; Tauxe, 2010).

5. Paleointensities

Obtaining reliable and high-quality estimates of the past strength of the Earth's magnetic field is notoriously difficult. Recently, de Groot et al. (2013) showed the benefits of using multiple paleointensity methods in the same study. This multimethod approach combines classical (double) heating methods with isothermal paleointensity methods, minimizing the effects of thermal alteration, and dramatically increasing the success rate to obtain reliable paleointensity estimates from lavas. In this study, we applied three different paleointensity methods to our set of cooling units: thermal IZZI-Thellier (Tauxe & Staudigel, 2004; Yu et al., 2004), microwave Thellier (Hill & Shaw, 1999; Walton et al., 1993) reducing the amount of heat applied to the samples, and the calibrated pseudo-Thellier method (de Groot et al., 2013) performed at room temperature.

The principle for these three paleointensity methods is the same: the natural remanent magnetization (NRM) is compared to laboratory-induced magnetizations, these are imparted by known magnetic fields to assess the strength of the paleofield that imparted the NRM. The laboratory magnetizations are imparted thermally for the IZZI-Thellier technique, resulting in (partial) thermal remanent magnetizations ((p)TRMs). Microwave excitation is used to impart magnetization for the microwave Thellier technique, and for pseudo-Thellier strong asymmetric alternating magnetic fields impart anhysteretic remanent magnetizations (ARMs) in the sample.

5.1. Thermal Thellier

Based on the rock-magnetic analyses and the thermal demagnetization behavior, 12 out of 16 dated sites were chosen and initially five samples per site were subjected to the thermal IZZI-Thellier protocol (Tauxe & Staudigel, 2004; Yu et al., 2004)

Continuo

Table 2



including pTRM-checks (Coe, 1967), with a bias field of 60 μ T. The samples were divided into two groups and the temperature steps were chosen accordingly: samples from rock-magnetic group L* were placed in the low-temperature group, with smaller temperature steps <400 °C and a total of 15 steps; the high-temperature group underwent additional steps >400 °C and went up to temperatures of 640 °C with a total of 16 steps. For three sites an additional 2-4 samples were measured in a second run, including pTRM-checks and a bias field of 40 μ T. To ensure that the magnetic moments of the individual samples did not exceed the dynamic range of the 2G magnetometer, small samples (~20 mg) were glued in borosilicate glass cups using quartz wool and KaSil glue. Samples were heated in an ASC TD48-SC thermal demagnetizer. The orientation of each sample was optimized so that the angle between the NRM and the laboratory field was always <90°, to suppress multidomain effects and help produce linear Arai plots (Paterson et al., 2015). To discriminate between reliable and unreliable data, a multitude of selection criteria has been proposed. Here, we used six predefined sets of selection criteria: TTA and TTB (Leonhardt et al., 2004), SELCRIT2 (Biggin et al., 2007), PICRIT03 (Kissel & Laj, 2004), Class A (de Groot et al., 2014), and CCRIT (Cromwell et al., 2015). For the first four sets, we used the modified version described by Paterson et al. (2014). We added the curvature criteria as defined by Paterson (2011) to all sets of selection criteria that did not already include this parameter, with |k'| < 0.35. To make the interpretation as objective as possible, the IZZI-Thellier data were analyzed with the auto-interpreter function of the Thellier-GUI (v. 3.13) (Shaar & Tauxe, 2013). All data were assessed using the six different sets of selection criteria (Figure 3), if a specific interpretation satisfied several selection criteria, it was taken into account multiple times, thereby this interpretation received more weight in the final result. The interpretations from samples of the same site were averaged, and the final site result was only accepted if the average paleointensity was calculated from at least three samples and the standard deviation divided by the mean paleointensity is below 20 % (bold values in Table 2).

From the 69 measured samples, 39 passed at least one of the selection criteria sets resulting in a total of 2,202 possible interpretations. Furthermore, 385 interpretations of 25 samples passed TTA(mod), 550 interpretations of 55 samples passed modified TTB(mod), 444 interpretations from 27 samples passed modified PICRIT03(mod), 579 interpretations of 35 samples passed modified SELCRIT2(mod), 223 interpretations of 25 samples passed Class A, and only 21 interpretations from four samples passed CCRIT. After applying the site selection criteria, we obtained an IZZI-Thellier average for 6 out of the 12 measured sites (Table 2). For site PI06, two specimens from core PI06-X3 show very high paleointensity results, 94.8 \pm 6.3 μ T. The results from five samples from the same site but from different cores (PI06-X2 and PI06-X5) show paleointensity estimates of 44.1 \pm 6.2 μ T. Therefore, this site does not pass the site selection criteria.

5.2. Microwave Thellier

The measurement protocol in a microwave Thellier experiment is similar to the thermal Thellier technique: the original magnetization of the samples is stepwise replaced by a magnetization imparted in a known laboratory field. The technique used to magnetize and demagnetize the samples, however, is different. Microwaves are used to directly excite the magnetic spins in the samples, therefore the thermal energy to which the samples are exposed is relatively small compared to thermal Thellier techniques. This procedure may therefore reduce the effects of chemical alteration of the magnetic signals (Hill & Shaw, 1999). All measurements were conducted on the "Tristan" microwave system installed in the geomagnetic laboratory of the University of Liverpool (UK). In the microwave system, one single sample is processed before moving onto the next sample. This makes it possible to optimize the steps for each sample individually. First, one specimen from a site was stepwise demagnetized to obtain information on the demagnetization behavior, second, a sister specimen from the same site and core was subjected to an IZZI-Thellier paleointensity protocol, including pTRM-checks, with a bias field of 40 µT. To test the suitability of the samples to the microwave Thellier technique, samples from 10 different sites were subjected to a microwave Thellier experiment. For five of these sites the first, or first two, samples showed promising paleointensity estimates and additional samples were measured with a maximum of six samples per site. A total of 33 specimens were measured on the microwave system. The interpretation of the microwave data is done with the Thellier-GUI and the same six sets of selection criteria are used for the interpretation of the data.





Figure 3. Paleointensity results. Results of sites PI02, PI07, and PI16 for thermal Thellier (a, d, g), microwave Thellier (b, e, h), and pseudo-Thellier (c, f, i) experiments. The results passed different sets of selection criteria, the passed sets for the plotted interpretation are specified in each panel: TTA, ThellierTool A mod (Leonhardt et al., 2004); TTB, ThellierTool B mod (Leonhardt et al., 2004); PICRIT03-mod (Kissel & Laj, 2004); S, SELCRIT2-mod (Biggin et al., 2007); all as modified by Paterson et al. (2014); A, Class A (de Groot et al., 2014); and C, CCRIT (Cromwell et al., 2015). The green lines in both Arai plot and Zijderveld diagram represent the interpreted segments. The gray numbers indicate the steps used for the experiments, temperature (°C) for thermal Thellier, the average microwave integral (W s) for microwave Thellier, and alternating field strength in (mT) for pseudo-Thellier.



In total, 25 samples passed at least one of the selection criteria sets (Table 2). The interpretations of some samples show ambiguous results, although passing the selection criteria. For example, sample PI07-06A (Figure 3e) losses 80% of the NRM in the first four demagnetization steps, 68.1 W s, resulting in a paleointensity estimate of 185.9 μ T. The quick demagnetization of this sample and the overprint on the NRM might indicate an effect of lightning although the other samples of this site do not show signs of lightning. As this is the only sample measured for this site, the behavior cannot be compared to other samples. Our microwave Thellier results highlight the importance of applying site selection criteria to the obtained individual sample interpretations. Of the 10 sites measured on the microwave system, two sites, PI01 and PI16, passed the site selection criteria.

5.3. pseudo Thellier

In a pseudo-Thellier experiment, samples are subjected to strong AFs, thereby avoiding thermally induced alteration caused by heating samples in conventional paleointensity methods altogether. While the thermal and microwave Thellier methods are absolute paleointensity methods, the pseudo-Thellier method is a relative paleointensity method (Tauxe et al., 1995). For lavas, absolute estimates of the paleofield can be obtained after calibration of the pseudo-Arai slope (de Groot et al., 2015, 2013, 2016; Lerner et al., 2017; Paterson et al., 2016). As the behavior of the pseudo-Thellier technique depends on the grain size distribution (Yu et al., 2003), samples suitable for calibration should satisfy the grain size selection proposed by de Groot (2013), which is the field at which half of the maximum ARM is gained for a sample, the $B_{1/2ARM}$ value. In this study, we use the revised calibration formula from de Groot et al. (2016): $B_{abs} = 7.718 \times |pTh-slope| + 14.600 \ \mu\text{T}$, with $23 \le B_{1/2ARM} \le 63 \text{ mT}$. We use this calibration formula as it is based on natural remanent magnetizations rather that laboratory-induced thermal magnetizations as proposed by Paterson et al. (2016).

Samples from all sites were subjected to the pseudo-Thellier paleointensity experiment. Five to 11 specimens were first stepwise demagnetized using alternating fields up to 270 mT (Section 4). The same alternating field steps and a bias field of 40 μ T are used to impart partial anhysteretic remanent magnetizations (pARMs). Lastly, the acquired ARM was demagnetized using the same field steps to check the linearity of the NRM-demag vs. ARM-demag. The data were analyzed with paleointensity.org (Béguin et al., 2020). The auto-interpretation function was used with a set of selection criteria specially designed for pseudo-Thellier experiments, pTh-SCRIT (Béguin et al., 2020). The focus of this set is on the linearity of the best-fit lines in all three plots (Arai, ARM-ARM, and Demag-Demag plots) and on the selection of the characteristic remnant magnetization (ChRM), where all signs of overprints on the NRM have to be removed to pass the selection. From the 146 measured specimens, 65 specimens passed the $B_{1/2ARM}$ criteria. Of which seven sites hold realistic paleointensity estimates based on the pTh-SCRIT criteria set and the site selection criteria (Figure 3 and Table 2).

6. Discussion

6.1. Paleodirections

For all sites with oriented samples, a reliable paleodirection could be obtained (Table 2). The samples were taken over a maximum distance of 20 m to average out local field anomalies and to obtain a more reliable paleodirection (e.g., Baag et al., 1995; de Groot & de Groot, 2019; Valet & Soler, 1999). This sampling strategy, however, might result in a lower k value than is expected for lavas. Nevertheless, our obtained k values are well above 70. Each cooling unit was sampled at one location with the exception of the 1562–1564 AD historical lava flow, which was sampled at the northern (PI01) and southern (PI02) branches. The paleodirections of PI01 and PI02, however, do not share a common true mean direction, meaning that these sites correspond to different cooling units or one of the sites might have moved postcooling, although there was no evidence for this in the field. Site PI13, with a mean calibrated radiocarbon age of 1543 AD, does show a common true mean direction with site PI02 (Koymans et al., 2020; Tauxe, 2010). Therefore, the direction of site PI02 is preferred over the direction of site PI01 for the 1562–1564 AD flow. The age of the PI01 flow is consequently considered unknown.



6.2. Multimethod Paleointensity Data From Pico Island

Seventeen sites were subjected to two or more paleointensity approaches, of which 12 sites passed at least one of the protocols and corresponding sample and site statistics; i.e., a site success rate of 70%. The three paleointensity methods rely on comparing the NRM of the specimen with a laboratory-induced magnetization, the technique used to induce the laboratory magnetization is different for each of the methods. The consistency of the paleointensity outcomes from the multimethod paleointensity technique can be considered as reliability check (e.g., Biggin & Paterson, 2014; Böhnel et al., 2009; de Groot et al., 2013; Monster et al., 2015). Site PI05 passed both pseudo-Thellier and thermal Thellier paleointensity protocols. Site PI16 passed all three paleointensity protocols. In both cases, the obtained paleointensity estimates from the different protocols are outside each other's confidence intervals. For these sites, the consistency between different methods is low. The thermal Thellier results of site PI16 is 24 μ T higher than the obtained pseudo-Thellier. The microwave Thellier result for site PI16 (59.4 μ T) is in-between the pseudo-Thellier and thermal Thellier result of 51.8 and 75.8 μ T, respectively. For site PI05, the pseudo-Thellier result is 10 μ T higher than the obtained thermal Thellier estimate. As all interpretations pass our selection criteria and are technically equally correct, it is impossible to imply which method yields the correct estimate of the paleofield. Sites with close ages, e.g., PI02 and PI13 or PI04 and PI07, show similar paleointensity estimates, within error, obtained by different methods; the thermal Thellier result of PI02 (44.0 \pm 6.1 μ T) and the pseudo-Thellier result of PI13 (41.9 \pm 5.4 μ T), also the pseudo-Thellier result of PI04 (39.3 \pm 2.2 μ T) and the thermal Thellier result of PI07 (43.2 \pm 4.9 μ T). This results in a high consistency between the paleointensity estimates obtained from different protocols for these sites.

6.3. Full-Vector Record for the Azores

The obtained paleomagnetic data from Pico Island can now be compared to the results of the nearby islands of São Miguel (Di Chiara et al., 2012, 2014b) and Terceira (de Groot et al., 2016), and existing geomagnetic field models (Figure 4) (Jackson et al., 2000; Korte et al., 2011; Nilsson et al., 2014; Pavón-Carrasco et al., 2014; Thébault et al., 2015). For this comparison, all data are relocated (Noel & Batt, 1990) to a central point in the Azores Archipelago (38.5°N, 28°W), just northeast of the coast of Pico Island. Due to the maximum distance between the islands within the Archipelago the maximum relocation error is within 1° and 0.8 µT (Casas & Incoronato, 2007), although the expected relocation error is probably much lower. The directional data from the three islands show coherent results and coeval sites show similar directions within error. The overall trend of the data is covered by the geomagnetic models, although the high and low variations we find are obviously smoothed in the models. The data reveal a period of high declination between 600 and 800 AD and around 1600 AD which is not predicted by the geomagnetic field models. The most prominent mismatch of the geomagnetic field models and the data from the Azores is the inclination $low (\sim 34^{\circ})$ between ~ 900 and 1560 AD. This inclination low is obtained from seven sites from the islands of Pico and São Miguel spanning this time period (sites PI02, PI13, PI15, PI16, SM08, SM09, and SM10). The decline of the inclination could have started shortly after 780 AD, the first low inclination value sampled is from 934 AD (site PI16; 893-986 AD). This deviation from current geomagnetic field models amplifies the importance of the Azores Archipelago to constrain geomagnetic behavior in the mid-North Atlantic region.

The intensity record shows more scattered results, and the existing geomagnetic field models are generally within the data uncertainties with the exception of site PI19 around 1844 BC with the lowest paleointensity ($31.2 \pm 1.4 \mu$ T) from the Pico sites and the intensity high observed from two sites SM03 from São Miguel with a paleointensity of 93.0 ± 11.7 μ T around 590 BC and PI14 with a paleointensity of 74.0 ± 4.6 μ T around 870 BC. For these two high intensities, corresponding declination and inclination values agree well with the geomagnetic field models.

6.4. Paleosecular Variation at the Azores During the Past 2 kyr

Full-vector paleosecular variation (PSV) curves can be obtained at three levels: global, regional, and local. The global PSV curves can be obtained from archeomagnetic models e.g., CALS10k.2 (Constable et al., 2016), SHA.DIF.14k (Pavón-Carrasco et al., 2014), or pfm9k (Nilsson et al., 2014). Lodge and Holme (2009)





Figure 4. Full-vector geomagnetic field record for the Azores. The declination (a), inclination (b), and intensity (c) results are shown for Pico (orange, this study), Terceira (light blue, de Groot et al., 2016) and São Miguel (dark blue, Di Chiara et al., 2012, 2014b). Geomagnetic field models for Pico Island are presented by the colored lines: SHA.DIF.14k red lines (Pavón-Carrasco et al., 2014) including one standard deviation envelops; pfm9k pink line (Nilsson et al., 2014); CALS10k (Korte et al., 2011) including one standard deviation error envelops teal colored line; and the IGRF/gufm1 in black (Jackson et al., 2000; Thébault et al., 2015). The declination and inclination data from Pico are depicted as orange circles with the associated error bars calculated from the α_{95} values (following calculations in Suttie and Nilsson (2019)). In the declination (a) and inclination (b) panels, data of Terceira are squares and from São Miguel are triangles; in the paleointensity panel (c), the shape of the data point reflects the paleointensity method used (see inset). The vertical error bars are the associated one standard deviation confidence intervals. The associated age errors are presented in Table 1 and supporting information S5.



used an existing global model, CALS7K.2 by Korte and Constable (2005), as an a priori model and used five published PSV curves to make a model for Europe to generate PSV curves at different locations. Regional PSV curves are generated by regional paleomagnetic models covering regions of continental scale, e.g., the regional model for the geomagnetic field in Europe of Pavón-Carrasco et al. (2008) using spherical cap harmonic analysis. To obtain local PSV curves, the paleomagnetic data for a small region or country can be interpolated to obtain the behavior of the paleofield for such small regions. For these local PSV curves, the paleomagnetic data must be relocated to a common point. Over the last decades different approached have been used to develop local PSV curves. Some curves are produced using hierarchical Bayesian modeling (e.g., Gómez-Paccard et al., 2012; Lanos, 2004; Lanos et al., 2005), while Thébault and Gallet (2010) introduced a bootstrap approach used by, e.g., Genevey et al. (2013) and Molina-Cardín et al. (2018). These approaches to derive local PSV curves have in common that they require a dense data input from which outliers can be identified and filtered from the data set.

The Azores Archipelago is located in a remote region in the middle of the North Atlantic Ocean, with the Canary Islands at ~1,500 km and the Iberian Peninsula at ~1,700 km distance. Since we find considerable deviations from existing geomagnetic field models, especially in our inclination data (Figure 4) we restricted our data set to only include data from the Azores islands. All data with independent age constrains are included in our data set (supplementary information S5). Sites that are paleomagnetically dated in any of the previous Azores studies (de Groot et al., 2016; Di Chiara et al., 2012) were not considered. Site PI01 is not included in the data set as the paleomagnetic data of this flow shows inconsistent results when compared to other sites with coeval ages. As the data density is very low before 100 AD, we can only interpolate the past 2 kyr for our full-vector PSV curve. The available data are bootstrapped with a subsampling of 5,000 points within the uncertainty of the geomagnetic data and the (radiocarbon) age constrains. The calibration of radiocarbon age to calendar years is nonlinear as a result of changes in the rate of 14 C production in the upper atmosphere. The geomagnetic direction is parametrically sampled from the Fisher distribution of the site. The uncertainty of the intensity data is subsampled using a Gaussian probability function with two standard deviation. The age uncertainty is described by a probability density function as the distribution is non-Gaussian due to the radiocarbon calibration, obtained with the MatCal software (Lougheed & Obrochta, 2016). For each subsample set, we ran a cubic smoothing spline, and the final PSV curve results from the median of the 5,000 individual curves with 68% and 95% error envelopes (Figure 5; supporting information S6). In addition to the Azores data set, synthetic data from the gufm1 model and the IGRF model were used to constrain the field expression from 1800 AD to the present-day field.

The resulting PSV curves (Figure 5) describe the declination high around 600–700 and 1350–1650 AD and the inclination low around 900–1500 AD. For sites with coeval ages and differing paleomagnetic data, the smoothing spline fitted the average value, as all data was weighted the same. By repeating this average fit in the bootstrap, the estimated error envelop can result in a lower range than would be expected from the data. This is, e.g., the case for the inclination record around 1550 AD, where the error envelop is rather small compared to the data used around this age. When no paleomagnetic data are present the error envelops widen, e.g., in all three curves around 400–550 AD. The main observation from our new PSV curve is the sweep presented in the inclination record. This inclination low is not (profoundly) observed at the Canary Islands (e.g., de Groot et al., 2015; Kissel et al., 2015). Compared to European PSV curves (e.g., Batt et al., 2017; Molina-Cardín et al., 2018; Pavón-Carrasco et al., 2009) and the geomagnetic field models evaluated for the Azores Archipelago, again the inclination record stands out. The inclination record shows deviating behavior from 900 to ~1700 AD. This shows the importance of a PSV curve in a remote location like the Azores Archipelago. The resulting PSV curves presented here can now be used to date volcanic products with unknown age that are within the time frame of the PSV curve, i.e., the past 2 kyr.

6.5. Paleomagnetic Dating with PSV Curves From the Azores

Seven sites (PI01, PI10, PI17, PI18, PI20, PI21, and PI22) were sampled from lavas with unknown ages. For these lava flows, an age range estimate is provided by the geological map of Nunes (1999) (Figure 1 and Table 3). The lava flows sampled at sites PI10, PI17, and PI18 are within the 2 kyr range of the PSV curve; while





Figure 5. Full-vector paleosecular variation curves past 2 kyr for the Azores. The bootstrapped declination (a), inclination (b), and intensity (c) curves are shown with the 68% and 95% confidence error envelops. The data used to produce the curves are plotted in the appropriate graphs, symbols, and colors used are the same as for Figure 4; the thin colored lines present the geomagnetic field models for Pico Island. The non-Gaussian age error distributions are considered in the bootstrap.



Table 3

Paleomagnetic Dating of Lava Flows From Pico Island With Unknown Ages

	Geological		Most
Sites	map age range (AD)	Paleomagnetic dating ranges (AD)	probable age (AD)
PI01	1562-1564	[574–600]; [652]; [654–774]	730
PI10	950-1450	[1280–1440]	1309
PI17	950-1450	[469–574]	525
PI18	950-1450	[28]; [36–40]; [370–518] ; [1683–1686]	447
PI20	-805050	_	_
PI21	-550-950	[1184–1233]	1212
PI22	-3050550	_	_

All flows are stratigraphically or historically constrained to the geological map age range. These broad ranges were narrowed to the paleomagnetic dating ranges provided; if the paleomagnetic dating yielded two age ranges, the one with the most probable age is given in bold. The most probable age is derived from the peak in the combined probability density function plot.

site PI21 might be within the range of the PSV curve but can also be up to 550 years older; and sites PI20 and PI22 fall outside the dating range of the PSV curve. We used the Matlab routine by Pavón-Carrasco et al. (2011) to estimate the age of the sites within the PSV dating range (supporting information S7), using our newly obtained PSV curves as master curve. The declination and inclination of the sites (Table 2) is compared to the PSV curves for the Azores. Sites PI01 and PI18 were the only flows that passed one of the paleointensity methods and for these flows the paleodirection together with the paleointensity (Table 2) was used to date the lava flows. The combined probability density functions of declination and inclination (and intensity for PI01 and PI18) result in multiple possible age ranges per flow (Table 3; supporting information S7).

Site PI01 was thought to be sampled on the northern branch of the 1562–1564 AD historical lava flow. However, the paleomagnetic data of this flow are inconsistent with site PI02 sampled on the southern branch of the same 1562–1564 AD flow and also inconsistent with site PI13, a radiocarbon dated flow with a calibrated age of 1543 AD. We thus assume that site PI01 was not sampled from the 1562 to 1564 AD lava flow but rather from an older flow below the historical flow. The paleomagnetic dating of this flow results in an age interval between 654 and 774 AD, validating this assumption. According to the geological map, the lava flows sampled at sites PI10, PI17, and PI18 are within the same age rang-

es (950–1450 AD). Site PI10 was sampled on the western part of the island in the lower slopes of Pico volcano, whereas sites PI17 and PI18 were sampled on the eastern part of the island in the Planalto da Achada fissure zone. The paleomagnetic dating of site PI10 result in a possible age between 1280 and 1440 AD with the highest probability at 1309 AD. The low inclination of PI10 fits the inclination low around this age interval. The obtained inclinations of sites PI17 and PI18 are too high to fit the inclination low of the Azores at this time interval, with measured inclinations of 59.8° and 57.0°, respectively. The ages of these flows are therefore not within the expected age range from the geological map. For site PI17, the paleomagnetic dating results in a single interval between 469 and 574 AD. The dating of site PI18 result in four possible intervals. Sites PI17 and PI18 is between 370 and 518 AD. Site PI21 was sampled on the western part of the island in the southern flank of Pico volcano, the age range provided by the geological map is -550-950 AD and this lava flow should be older than the lava flow sampled at site PI10. The measured inclination low (~900–1560 AD). The combined probability of the inclination and declination result in a age interval of 1184–1233 AD.

7. Conclusions

The volcanic products from Pico Island, in the Azores Archipelago, proved to be good recorders of geomagnetic field variations in the mid-North Atlantic region. We obtained paleointensity values for 12 of the 22 measured sites, of which 10 sites with known age. By combing our new data with existing data for the neighboring islands of São Miguel and Terceira we compiled the first full-vector PSV curve for the Azores region for the last 2,000 years. By using a bootstrap method, the non-Gaussian age distribution resulting from the calibration of radiocarbon ages was considered, together with the uncertainty of the paleomagnetic data. The data and PSV curves reveal an inclination low from ~900 to 1560 AD, with minimum values of 32°. This inclination low is not previously constrained by global geomagnetic models and highlights the importance of a PSV curve for the Azores region. Furthermore, we illustrated the potential of paleomagnetic dating using our new full-vector PSV curve for the Azores by successfully constraining the ages of five lava flows from Pico Island.



Acknowledgments

Liz van Grinsven, Anna Gülcher, and Flor Wassing are gratefully acknowledged for their assistance in the field and carrying out parts of the measurements under our supervision as part of their BSc graduation projects. Matthijs Bloem is also gratefully acknowledged for preparing the SEM images and carrying out the thermal Thellier measurements under our supervision for his MSc graduation project. Andrew J. Biggin is acknowledged for the supervision on the microwave Thellier experiments. Wout Krijgsman is thanked for his help in the field. We thank J. C. Nunes for providing information about the radiocarbon ages. L. V. de Groot acknowledges NWO VENI grant 863.15.003. We thank F. Javier Pavón-Carrasco and Anita Di Chiara for their constructive comments that significantly improved the manuscript and PSV curve. The paleomagnetism.org file with the directional data can be found in the supporting information (S4). The data used for the construction of the PSV curves are provided as table in the supporting information (S5). The PSV curves can be found in the supporting information (S8). All new geomagnetic data for Pico Island can be found in the MagIC database repository at https:// earthref.org/MagIC/16776.

References

- Baag, C., Helsley, C. E., Xu, S., & Lienert, B. R. (1995). Deflection of paleomagnetic directions due to magnetization of the underlying terrain. Journal of Geophysical Research, 100(B6), 10013–10027. https://doi.org/10.1029/95JB00148
- Batt, C. M., Brown, M. C., Clelland, S.-J., Korte, M., Linford, P., & Outram, Z. (2017). Advances in archaeomagnetic dating in Britain: New data, new approaches and a new calibration curve. *Journal of Archaeological Science*, 85, 66–82. https://doi.org/10.1016/j.jas.2017.07.002
- Béguin, A., Paterson, G. A., Biggin, A. J., & Groot, L. V. (2020). Paleointensity.org: An online, open source, application for the interpretation of paleointensity data. *Geochemistry, Geophysics, Geosystems*, 21, e2019GC008791. https://doi.org/10.1029/2019GC008791
- Biggin, A. J., & Paterson, G. A. (2014). A new set of qualitative reliability criteria to aid inferences on palaeomagnetic dipole moment variations through geological time. Frontiers in Earth Science, 2, 24. https://doi.org/10.3389/feart.2014.00024
- Biggin, A. J., Perrin, M., & Dekkers, M. J. (2007). A reliable absolute palaeointensity determination obtained from a non-ideal recorder. *Earth and Planetary Science Letters*, 257(3–4), 545–563. https://doi.org/10.1016/j.epsl.2007.03.017
- Böhnel, H., Pavón-Carrasco, F. J., Sieron, K., & Mahgoub, A. N. (2016). Palaeomagnetic dating of two recent lava flows from Ceboruco volcano, western Mexico. *Geophysical Journal International*, 207(2), 1203–1215. https://doi.org/10.1093/gji/ggw310Refstyled
- Böhnel, H. N., Dekkers, M. J., Delgado-Argote, L. A., & Gratton, M. N. (2009). Comparison between the microwave and multispecimen parallel difference pTRM paleointensity methods. *Geophysical Journal International*, 177(2), 383–394. https://doi. org/10.1111/j.1365-246X.2008.04036.x
- Casas, L., & Incoronato, A. (2007). Distribution analysis of errors due to relocation of geomagnetic data using the 'Conversion via Pole' (CVP) method: Implications on archaeomagnetic data. *Geophysical Journal International*, 169(2), 448–454. https://doi. org/10.1111/j.1365-246X.2007.03346.x
- Coe, R. S. (1967). Paleo-intensities of the Earth's magnetic field determined from Tertiary and Quaternary rocks. *Journal of Geophysical Research*, 72(12), 3247–3262. https://doi.org/10.1029/JZ072i012p03247
- Constable, C., Korte, M., & Panovska, S. (2016). Persistent high paleosecular variation activity in southern hemisphere for at least 10 000 years. *Earth and Planetary Science Letters*, 453, 78–86. https://doi.org/10.1016/j.epsl.2016.08.015
- Costa, A. C. G., Marques, F. O., Hildenbrand, A., Sibrant, A. L. R., & Catita, C. M. S. (2014). Large-scale catastrophic flank collapses in a steep volcanic ridge: The Pico–Faial Ridge, Azores Triple Junction. *Journal of Volcanology and Geothermal Research*, 272, 111–125. https://doi.org/10.1016/j.jvolgeores.2014.01.002
- Cromwell, G., Tauxe, L., & Halldórsson, S. A. (2015). New paleointensity results from rapidly cooled Icelandic lavas: Implications for Arctic geomagnetic field strength. Journal of Geophysical Research: Solid Earth, 120, 2913–2934. https://doi.org/10.1002/2014JB011828
- de Groot, B. M., & de Groot, L. V. (2019). A low-cost device for measuring local magnetic anomalies in volcanic terrain. Geoscientific Instrumentation, Methods and Data Systems, 8(2), 217–225. https://doi.org/10.5194/gi-8-217-2019
- de Groot, L. V., Béguin, A., Kosters, M. E., van Rijsingen, E. M., Struijk, E. L. M., Biggin, A. J., et al. (2015). High paleointensities for the Canary Islands constrain the Levant geomagnetic high. *Earth and Planetary Science Letters*, 419, 154–167. https://doi.org/10.1016/j. epsl.2015.03.020
- de Groot, L. V., Biggin, A. J., Dekkers, M. J., Langereis, C. G., & Herrero-Bervera, E. (2013). Rapid regional perturbations to the recent global geomagnetic decay revealed by a new Hawaiian record. Nature Communications, 4(1), 2727. https://doi.org/10.1038/ncomms3727
- de Groot, L. V., Dekkers, M. J., Visscher, M., & ter Maat, G. W. (2014). Magnetic properties and paleointensities as function of depth in a Hawaiian lava flow. *Geochemistry, Geophysics, Geosystems*, 15, 1096–1112. https://doi.org/10.1002/2013GC005094
- de Groot, L. V., Pimentel, A., & Di Chiara, A. (2016). The multimethod palaeointensity approach applied to volcanics from Terceira: Full-vector geomagnetic data for the past 50 kyr. *Geophysical Journal International*, 206(1), 590–604. https://doi.org/10.1093/gji/ggw095
 Demande, J., Fabriol, R., Gérard, A., & Iundt, F. (1982). *Prospection géothermique des Iles de Faial et Pico (Açores)*. Orléans: Bureau de Recherches Géologiques et Minières.
- Di Chiara, A., Speranza, F., & Porreca, M. (2012). Paleomagnetic secular variation at the Azores during the last 3 ka. Journal of Geophysical Research, 117, B07101. https://doi.org/10.1029/2012JB009285
- Di Chiara, A., Speranza, F., Porreca, M., Pimentel, A., D'Ajello Caracciolo, F., & Pacheco, J. (2014a). Constraining chronology and timespace evolution of Holocene volcanic activity on the Capelo Peninsula (Faial Island, Azores): The paleomagnetic contribution. *Bulletin* of the Geological Society of America, 126(9–10), 1164–1180. https://doi.org/10.1130/B30933.1
- Di Chiara, A., Tauxe, L., & Speranza, F. (2014b). Paleointensity determination from São Miguel (Azores Archipelago) over the last 3ka. *Physics of the Earth and Planetary Interiors*, 234(C), 1–13. https://doi.org/10.1016/j.pepi.2014.06.008
- Dunlop, D. J. (2002). Theory and application of the Day plot (M_{rs}/M_s versus H_{cr}/H_c) 1. Theoretical curves and tests using titanomagnetite data. Journal of Geophysical Research, 107(B3), 2056. https://doi.org/10.1029/2001JB000486
- Fisher, R. A. (1953). Dispersion on a sphere. Proceedings of the Royal Society of London-Series A: Mathematical and Physical Sciences, 217(1130), 295–305. https://doi.org/10.1098/rspa.1953.0064
- Gaspar, J. L., Guest, J. E., Duncan, A. M., Barriga, F. J. A. S., & Chester, D. K. (2015). Volcanic geology of São Miguel island (Azores archipelago).. London, UK: Geological Society.
- Genevey, A., Gallet, Y., Thébault, E., Jesset, S., & Le Goff, M. (2013). Geomagnetic field intensity variations in Western Europe over the past 1100 years. Geochemistry, Geophysics, Geosystems, 14, 2858–2872. https://doi.org/10.1002/ggge.20165
- Gómez-Paccard, M., Chauvin, A., Lanos, P., Dufresne, P., Kovacheva, M., Hill, M. J., et al. (2012). Improving our knowledge of rapid geomagnetic field intensity changes observed in Europe between 200 and 1400 AD. Earth and Planetary Science Letters, 355(356), 131–143. https://doi.org/10.1016/j.epsl.2012.08.037
- Greve, A., Turner, G. M., Conway, C. E., Townsend, D. B., Gamble, J. A., & Leonard, G. S. (2016). Palaeomagnetic refinement of the eruption ages of Holocene lava flows, and implications for the eruptive history of the Tongariro Volcanic Centre, New Zealand. *Geophysical Journal International*, 207(2), 702–718. https://doi.org/10.1093/gji/ggw296
- Hill, M. J., & Shaw, J. (1999). Palaeointensity results for historic lavas from Mt Etna using microwave demagnetization/remagnetization in a modified Thellier-type experiment. *Geophysical Journal International*, 139(2), 583–590. https://doi.org/10.1046/j.1365-246X.1999.00980.x
- Jackson, A., Jonkers, A. R. T., & Walker, M. R. (2000). Four centuries of geomagnetic secular variation from historical records. *Philosophical Transactions of the Royal Society of London, Series A: Mathematical, Physical and Engineering Sciences, 358*(1768), 957–990. https://doi.org/10.1098/rsta.2000.0569
- Johnson, C. L., Wijbrans, J. R., Constable, C. G., Gee, J., Staudigel, H., Tauxe, L., et al. (1998). 40Ar/39Ar ages and paleomagnetism of São Miguel lavas, Azores. Earth and Planetary Science Letters, 160(3), 637–649. https://doi.org/10.1016/S0012-821X(98)00117-4



Kissel, C., & Laj, C. (2004). Improvements in procedure and paleointensity selection criteria (PICRIT-03) for Thellier and Thellier determinations: Application to Hawaiian basaltic long cores. *Physics of the Earth and Planetary Interiors*, 147(2–3), 155–169. https://doi.org/10.1016/j.pepi.2004.06.010

Kissel, C., Rodriguez-Gonzalez, A., Laj, C., Perez-Torrado, F., Carracedo, J. C., Wandres, C., & Guillou, H. (2015). Paleosecular variation of the earth magnetic field at the Canary Islands over the last 15 ka. *Earth and Planetary Science Letters*, 412(C), 52–60. http://dx.doi. org/10.1016/j.epsl.2014.12.031

- Korte, M., & Constable, C. G. (2005). Continuous geomagnetic field models for the past 7 millennia: 2. CALS7K. Geochemistry, Geophysics, Geosystems, 6, Q02H16. https://doi.org/10.1029/2004GC000801
- Korte, M., Constable, C., Donadini, F., & Holme, R. (2011). Reconstructing the Holocene geomagnetic field. Earth and Planetary Science Letters, 312(3), 497–505. https://doi.org/10.1016/j.epsl.2011.10.031
- Koymans, M. R., Langereis, C. G., Pastor-Galán, D., & van Hinsbergen, D. J. J. (2016). Paleomagnetism.org: An online multi-platform open source environment for paleomagnetic data analysis. *Computers and Geosciences*, 93(C), 127–137. https://doi.org/10.1016/j. cageo.2016.05.007
- Koymans, M. R., van Hinsbergen, D. J. J., Pastor-Galán, D., Vaes, B., & Langereis, C. G. (2020). Towards FAIR paleomagnetic data management through paleomagnetism.org 2.0. *Geochemistry, Geophysics, Geosystems*, 21, e2019GC008838. https://doi.org/10.1029/2019GC008838
- Lanos, P. (2004). Tools for constructing chronologies. In C. E. Buck, & A. R. Millard (Eds.), Tools for constructing chronologies: Crossing disciplinary boundaries (Vol. 177). London: Springer. https://doi.org/10.1007/978-1-4471-0231-1
- Lanos, P., Le Goff, M., Kovacheva, M., & Schnepp, E. (2005). Hierarchical modelling of archaeomagnetic data and curve estimation by moving average technique. *Geophysical Journal International*, *160*(2), 440–476. https://doi.org/10.1111/j.1365-246X.2005.02490.x
- Leonhardt, R., Heunemann, C., & Krása, D. (2004). Analyzing absolute paleointensity determinations: Acceptance criteria and the software ThellierTool4.0. Geochemistry, Geophysics, Geosystems, 5, Q12016. https://doi.org/10.1029/2004GC000807
- Lerner, G. A., Smirnov, A. V., Surovitckii, L. V., & Piispa, E. J. (2017). Nonheating methods for absolute paleointensity determination: Comparison and calibration using synthetic and natural magnetite-bearing samples. *Journal of Geophysical Research: Solid Earth*, 122, 1614–1633. https://doi.org/10.1002/2016JB013777
- Lodge, A., & Holme, R. (2009). Towards a new approach to archaeomagnetic dating in Europe using geomagnetic field modelling. Archaeometry, 51(2), 309–322. https://doi.org/10.1111/j.1475-4754.2008.00400.x
- Lougheed, B. C., & Obrochta, S. P. (2016). MatCal: Open source Bayesian 14C age calibration in Matlab. *Journal of Open Research Software*, 4, e42. https://doi.org/10.5334/jors.130
- Molina-Cardín, A., Campuzano, S. A., Osete, M. L., Rivero-Montero, M., Pavón-Carrasco, F. J., Palencia-Ortas, A., et al. (2018). Updated Iberian archeomagnetic catalogue: New full vector paleosecular variation curve for the last three millennia. *Geochemistry, Geophysics, Geosystems*, 19, 3637–3656. https://doi.org/10.1029/2018GC007781
- Monster, M. W. L., de Groot, L. V., Biggin, A. J., & Dekkers, M. J. (2015). The performance of various palaeointensity techniques as a function of rock magnetic behaviour—A case study for La Palma. *Physics of the Earth and Planetary Interiors*, 242, 36–49. https://doi.org/10.1016/j.pepi.2015.03.004
- Mullender, T. A. T., Frederichs, T., Hilgenfeldt, C., de Groot, L. V., Fabian, K., & Dekkers, M. J. (2016). Automated paleomagnetic and rock magnetic data acquisition with an in-line horizontal "2G" system. *Geochemistry, Geophysics, Geosystems*, 17, 3546–3559. https://doi. org/10.1002/2016GC006436
- Mullender, T. A. T., Velzen, A. J., & Dekkers, M. J. (1993). Continuous drift correction and separate identification of ferrimagnetic and paramagnetic contributions in thermomagnetic runs. *Geophysical Journal International*, 114(3), 663–672. https://doi.org/10.1111/j.1365-246X.1993.tb06995.x
- Nilsson, A., Holme, R., Korte, M., Suttie, N., & Hill, M. (2014). Reconstructing Holocene geomagnetic field variation: New methods, models and implications. *Geophysical Journal International*, 198(1), 229–248. https://doi.org/10.1093/gji/ggu120
- Noel, M., & Batt, C. M. (1990). A method for correcting geographically separated remanence directions for the purpose of archaeomagnetic dating. *Geophysical Journal International*, 102(3), 753–756. https://doi.org/10.1111/j.1365-246X.1990.tb04594.x
- Nunes, J. C. (1999). A actividade vulcânica na ilha do Pico do Plistocénico Superior ao Holocénico: Mecanismo eruptivo e hazard vulcânico (PhD thesis). Ponta Delgada: University of Azores.
- Nunes, J. C. (2020). Geology and volcanology of Pico Island (Azores, Portugal): A field guide. In F. Fernandes, A. Malheiro, & H. I. Chaminé (Eds.), Advances in natural hazards and hydrological risks: Meeting the challenge (pp. 183–192). Cham: Springer International Publishing. https://doi.org/10.1007/978-3-030-34397-2_35
- Paterson, G. A. (2011). A simple test for the presence of multidomain behavior during paleointensity experiments. *Journal of Geophysical Research*, *116*, B10104. https://doi.org/10.1029/2011JB008369
- Paterson, G. A., Biggin, A. J., Hodgson, E., & Hill, M. J. (2015). Thellier-type paleointensity data from multidomain specimens. Physics of the Earth and Planetary Interiors, 245, 117–133. https://doi.org/10.1016/j.pepi.2015.06.003
- Paterson, G. A., Heslop, D., & Pan, Y. (2016). The pseudo-Thellier palaeointensity method: New calibration and uncertainty estimates. Geophysical Journal International, 207(3), 1596–1608. https://doi.org/10.1093/gji/ggw349
- Paterson, G. A., Tauxe, L., Biggin, A. J., Shaar, R., & Jonestrask, L. C. (2014). On improving the selection of Thellier-type paleointensity data. Geochemistry, Geophysics, Geosystems, 15, 1180–1192. https://doi.org/10.1002/2013GC005135
- Pavón-Carrasco, F. J., Osete, M. L., Torta, J. M., & De Santis, A. (2014). A geomagnetic field model for the Holocene based on archaeomagnetic and lava flow data. Earth and Planetary Science Letters, 388, 98–109. https://doi.org/10.1016/j.epsl.2013.11.046
- Pavón-Carrasco, F. J., Osete, M. L., Torta, J. M., & Gaya-Piqué, L. R. (2009). A regional archeomagnetic model for Europe for the last 3000 years, SCHA.DIF.3K: Applications to archeomagnetic dating. *Geochemistry, Geophysics, Geosystems, 10*, Q03013. https://doi. org/10.1029/2008GC002244
- Pavón-Carrasco, F. J., Osete, M. L., Torta, J. M., Gaya-Piqué, L. R., & Lanos, P. (2008). Initial SCHA.DI.00 regional archaeomagnetic model for Europe for the last 2000 years. *Physics and Chemistry of the Earth, Parts A/B/C*, 33(6–7), 596–608. https://doi.org/10.1016/j. pce.2008.02.024
- Pavón-Carrasco, F. J., Rodríguez-González, J., Osete, M. L., & Torta, J. M. (2011). A Matlab tool for archaeomagnetic dating. Journal of Archaeological Science, 38(2), 408–419. https://doi.org/10.1016/j.jas.2010.09.021
- Reimer, P. J., Bard, E., Bayliss, A., Beck, J. W., Blackwell, P. G., Ramsey, C. B., et al. (2013). IntCal13 and Marine13 radiocarbon age calibration curves 0–50,000 years cal BP. Radiocarbon, 55(4), 1869–1887. https://doi.org/10.2458/azu_js_rc.55.16947
- Shaar, R., & Tauxe, L. (2013). Thellier GUI: An integrated tool for analyzing paleointensity data from Thellier-type experiments. Geochemistry, Geophysics, Geosystems, 14, 677–692. https://doi.org/10.1002/ggge.20062



- Suttie, N., & Nilsson, A. (2019). Archaeomagnetic data: The propagation of an error. *Physics of the Earth and Planetary Interiors*, 289, 73–74. https://doi.org/10.1016/j.pepi.2019.02.008
- Tauxe, L. (2010). Essentials of paleomagnetism. Berkeley, CA: University of California Press. https://doi.org/10.1525/9780520946378
- Tauxe, L., Pick, T., & Kok, Y. S. (1995). Relative paleointensity in sediments: A pseudo-Thellier approach. Geophysical Research Letters, 22(21), 2885–2888. https://doi.org/10.1029/95GL03166
- Tauxe, L., & Staudigel, H. (2004). Strength of the geomagnetic field in the Cretaceous normal Superchron: New data from submarine basaltic glass of the Troodos Ophiolite. *Geochemistry, Geophysics, Geosystems, 5*, Q02H06. https://doi.org/10.1029/2003GC000635
- Thébault, E., Finlay, C. C., Beggan, C. D., Alken, P., Aubert, J., Barrois, O., et al. (2015). International geomagnetic reference field: The 12th generation. *Earth, Planets and Space*, 67(1), 79. https://doi.org/10.1186/s40623-015-0228-9
- Thébault, E., & Gallet, Y. (2010). A bootstrap algorithm for deriving the archeomagnetic field intensity variation curve in the Middle East over the past 4 millennia BC. *Geophysical Research Letters*, *37*, L22303. https://doi.org/10.1029/2010GL044788
- Valet, J.-P., & Soler, V. (1999). Magnetic anomalies of lava fields in the Canary Islands. Possible consequences for paleomagnetic records. *Physics of the Earth and Planetary Interiors*, 115(2), 109–118. https://doi.org/10.1016/S0031-9201(99)00071-0
- Vandamme, D. (1994). A new method to determine paleosecular variation. *Physics of the Earth and Planetary Interiors*, 85(1-2), 131-142. https://doi.org/10.1016/0031-9201(94)90012-4
- Walton, D., Share, J., Rolph, T. C., & Shaw, J. (1993). Microwave magnetisation. Geophysical Research Letters, 20(2), 109–111. https://doi.org/10.1029/92GL02782
- Watkins, N. D., & Haggerty, S. E. (1967). Primary oxidation variation and petrogenesis in a single lava. Contributions to Mineralogy and Petrology, 15(3), 251–271. https://doi.org/10.1007/BF01185345
- Yu, Y., Dunlop, D. J., & Özdemir, Ö. (2003). Are ARM and TRM analogs? Thellier analysis of ARM and pseudo-Thellier analysis of TRM. Earth and Planetary Science Letters, 205(3–4), 325–336. https://doi.org/10.1016/S0012-821X(02)01060-9
- Yu, Y., Tauxe, L., & Genevey, A. (2004). Toward an optimal geomagnetic field intensity determination technique. Geochemistry, Geophysics, Geosystems, 5, Q02H07. https://doi.org/10.1029/2003GC000630
- Zanon, V., Pimentel, A., Auxerre, M., Marchini, G., & Stuart, F. M. (2020). Unravelling the magma feeding system of a young basaltic oceanic volcano. *Lithos*, 352–353, 105325. https://doi.org/10.1016/j.lithos.2019.105325